Author's response to reviews

Title: A microRNA profile associated with Opisthorchis viverrini-induced cholangiocarcinoma in tissue and plasma.

Authors:

    Jordan Plieskatt (jordanp@gwu.edu)
    Gabriel Rinaldi (grinaldi@gwu.edu)
    Yanjun Feng (fengyajun.fyi@gmail.com)
    Jin Peng (pj_2011@gwmail.gwu.edu)
    Samantha Easley (seasley@mfa.gwu.edu)
    Xinying Jia (sidney.jia@qimrberghofer.edu.au)
    Jeremy Potriquet (jeremy.potriquet@qimrberghofer.edu.au)
    Chawalit Pairojkul (chawal-p@kku.ac.th)
    Vajarabhongsa Bhudhisawasdi (yoevajara@gmail.com)
    Banchob Sripa (banchob@kku.ac.th)
    Paul J Brindley (pbrindley@email.gwu.edu)
    Jeffrey Bethony (jbethony@gwu.edu)
    Jason Mulvenna (jason.mulvenna@qimrberghofer.edu.au)

Version: 5
Date: 20 March 2015

Author's response to reviews: see over
Dear Dr Budczies:

Thank you for efforts in regard to our manuscript, “A microRNA profile associated with Opisthorchis viverrini-induced cholangiocarcinoma in tissue and plasma” (MS: 2032896589150592). We also thank the reviewers for their valuable comments and would like to submit a revised manuscript for your consideration. We have provided a point-by-point response to the latest comments as follows and have uploaded a revised manuscript:

1. **Please add a Competing Interest section**
   None of the authors have a competing interest and we have declared this on Page 490 of the revised manuscript.

Kind regards,

Jason P Mulvenna
Infectious Disease and Cancer
QIMR Berghofer Medical Research Institute
Herston Rd
Brisbane, Australia, 4006